Author's response to reviews

Title: Engaging participants in a complex intervention trial in Australian General Practice

Authors:

David Perkins (dperkins@gwahs.health.nsw.gov.au)
Mark F Harris (m.f.harris@unsw.edu.au)
Jocelyn Tan (j.tan@unsw.edu.au)
Bettina Christl (b.christl@unsw.edu.au)
Jane Taggart (j.taggart@unsw.edu.au)
Mahnaz Fanaian (m.fanaian@unsw.edu.au)

Version: 9 Date: 11 August 2008

Author's response to reviews:

Dear Editor
As requested, we have revised:
Section headings
Abbreviations
Section heading 'Authors' contributions'
Added all authors to the reference list
Checked the manuscript
Kind regards
Mark Harris for the authors